Author's response to reviews

Title: What do we know about the non-work determinants of workers' mental health? A systematic review of longitudinal studies

Authors:

Nancy Beauregard (nancy.beauregard.2@umontreal.ca)
Alain Marchand (alain.marchand@umontreal.ca)
Marie-Ève Blanc (marie-eve.blanc@umontreal.ca)

Version: 5 Date: 19 May 2011

Author's response to reviews:

Object: Minor revisions, Manuscript ID 6834217924988574

Hello,

Few minor errors were detected with the latest version of the article. Essentially:

1. Main text
   p.4 : the word "our" was omitted to make the sentence neutral.
   p.6. the words "factors" were changed for "determinants" for consistency in terminology.
   p.6. the number 1) preceding the research question was omitted.
   p.7. the word "society" was added for consistency in terminology.
   p.7. the expression "chronic health conditions" was corrected.
   p.11. typographic error "work-exposure".
   p.16. typographic error "family- and network-level".
   p.19. the word "determinants' was changed for "factors' for consistency in terminology.

2. Tables
   p.40 (T-1). the expression "men=ns" was doubled.
   p.46 (T-3). the word "or' was changed for 'and'.

Thank you for considering these last revisions.

Regards,